Author’s response to reviews

Title: The chemotactic properties of various topical brimonidine tartrate ophthalmic preparations

Authors:

RUIZ ALONSO (ruizsa@hotmail.com)

HELENA SOLARI (hpsolari74@gmail.com)

EDUARDO DAMASCENO (e_damasceno@yahoo.com)

MIGUEL BURNIER (mburnier@yahoo.com)

MARCELO VENTURA (palis@infolink.com.br)

Version: 4 Date: 26 Feb 2020

Author’s response to reviews:

Editor Comments:

1. Thank you for clarifying the consent in the requested revision. Please now add this statement regarding consent obtained into the 'Ethics approval and consent to participate section'.

R: Added.

2. Please rename Graphic 1 to Figure 2, and ensure Graphic 1 is now cited as Figure 2 in the text and legend.

R: Renamed in the text and legend.

3. Please remove the editing certificate and editor letter from the manuscript, as at this stage they are no longer necessary.

R: Removed.

4. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional-supplementary files are cited within the text

R: Fixed.
Thank you for your consideration. I look forward to hearing from you.

Sincerely,
Ruiz Simonato Alonso.
Universidade Federal Fluminense.
Ophthalmology Department.

ruizsa@hotmail.com